Author's response to reviews

Title: Predictors of Adherence to Anti Retro Viral Therapy (ART) among adult people living with HIV and attending their clinical care, Eastern Ethiopia

Authors:

Shiferaw - Letta (samishife2006@yahoo.com)
Asrat - Demissie (asrat_dem@yahoo.com)
Lemessa - Oljira (Olemessa@yahoo.com)
Yadeta - Dessie (yad_de2005@yahoo.com)

Version: 4 Date: 8 December 2014

Author's response to reviews:

Date 08/12/2014
To: BMC International Health and Human Rights
Subject: Submission of Manuscript for publication
From: Shiferaw Letta, Asrat Demissie, Lemessa Oljira, Yadeta Dessie

We would like to submit our original research manuscript entitled Predictors of Adherence to Anti Retro Viral Therapy (ART) among adult people living with HIV and attending their clinical care, Eastern Ethiopia. The research assessed Predictors of Adherence to Anti Retro Viral Therapy in a wider scale in 'Khat' chewing area of the eastern part of the country. We believe that the findings identified in this study would add important knowledge in the area HIV treatment and care in low income country and as well the local context.

We have chosen your journal as the fact that our research finding would reach the right audiences as fast as possible in cognizant of the potential wider scope of the journal readers, the journal scope also include our research topic, high quality peer review and rapidity of your publication process.

Here, we declared that this research is our original work, has never been previously published and is not being considered for publication elsewhere and all sources have been duly acknowledged.

Regards,
Shiferaw Letta, the corresponding author